Author’s response to reviews

Title: Magnetic Resonance Imaging Findings of Idiopathic Granulomatous Mastitis: Can it be an Indirect Sign of Treatment Success or Fail?

Authors:

Ayşegül Altunkeser (aaltunkeser@hotmail.com)
Zeynep Arslan (zeynep_a1002@hotmail.com)
Mehmet Eryılmaz (malieryilmaz@hotmail.com)

Version: 3 Date: 14 Nov 2019

Author’s response to reviews:

Author's response to review

BMIM-D-18-00246R2

Magnetic Resonance Imaging Findings of Idiopathic Granulomatous Mastitis: Can it be an Indirect Sign of Treatment Success or Fail?

BMC Medical Imaging

Authors:

Ayşegül Altunkeser(aaltunkeser@hotmail.com)
Zeynep Fatma Arslan: zeynep_a1002@hotmail.com
Mehmet Ali Eryılmaz: mali_eryilmaz@hotmail.com

Version: 3 Date: 1 Nov 2019

Author's response to review: see over
REQUESTED REVISIONS:

1. I just suggest to describe what happened to the 17 patients who did not have treatment success: what was the BIRADS assessments? How many were symptomatic? Did they have follow-up? How many performed surgery or any additional treatment?

17 patients who failed treatment were categorized as BI-RADS 3. Symptoms of 7 patients regressed spontaneously in the follow-up period. Also, one patient developed fistul and than recovered. The remaining 9 patients had pain and swelling. Sonography showed abscess in 5 patients and fistula tract in 1 patient. One of these patients underwent abscess drainage but treatment failed. These sentences were added to the results section. Page 5, line 131

We did the revisions that you suggested. I believe my manuscript is more valuable for publication in the journal in the light of advices of you.